codex alimentarius commission



FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS WORLD HEALTH ORGANIZATION



JOINT OFFICE: Viale delle Terme di Caracalla 00100 ROME Tel: 39 06 57051 www.codexalimentarius.net Email: codex@fao.org Facsimile: 39 06 5705 4593

ALINORM 01/14

JOINT FAO/WHO FOOD STANDARDS PROGRAMME

CODEX ALIMENTARIUS COMMISSION

Twenty-fourth Session, Geneva, 2-7 July 2001

REPORT OF THE EIGHTEENTH SESSION OF THE CODEX COMMITTEE ON COCOA PRODUCTS AND CHOCOLATE Fribourg, Switzerland, 2-4 November 2000

Note: This report contains the Circular Letter CL 2000/46-CPC

codex alimentarius commission

FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS WORLD HEALTH ORGANIZATION

JOINT OFFICE: Viale delle Terme di Caracalla 00100 ROME Tel.: 57051 Telex: 625825-625853 FAO I E-mail: Codex@fao.org Facsimile: +39(06)5705.4593

CX 5/1.2

CL 2000/46-CPC December 2000

- TO: Codex Contact Points Interested International Organizations
- **FROM:** Secretary, Joint FAO/WHO Food Standards Programme FAO, 00100 Rome, Italy
- **SUBJECT: DISTRIBUTION OF THE REPORT OF THE EIGHTEENTH SESSION OF THE CODEX COMMITTEE ON CHOCOLATE AND CHOCOLATE PRODUCTS (ALINORM 01/14)** The report of the Eighteenth Session of the Codex Committee on Chocolate and Chocolate Products will be considered by the 24th Session of the Codex Alimentarius Commission (Geneva, 2-7 July 2001)
 - PART A MATTERS FOR ADOPTION BY THE TWENTY FOURTH SESSION OF THE CODEX ALIMENTARIUS COMMISSION AT STEP 8
 - 1. DRAFT STANDARD FOR COCOA BUTTER AT STEP 8 (ALINORM 01/14 - PARAS 7-16 AND APPENDIX II)

Governments and interested international organizations are invited to comment on the above standard and should do so in writing in conformity with the Guide to the Consideration of Standards at Step 8 (See *Procedural Manual of the Codex Alimentarius Commission*, Eleventh Edition, page 26-27) to the Secretary of the Codex Alimentarius Commission, Joint FAO/WHO Food Standards Programme, FAO, Viale delle Terme di Caracalla,00100 Rome, Italy, Fax: +39 06570 54593, Email: Codex@fao.org before 1st February 2001

2. DRAFT STANDARD FOR COCOA (CACAO) MASS (COCOA/CHOCOLATE LIQUOR) AND COCOA CAKE AT STEP 8 (ALINORM 01/14 - PARAS 17-27 AND APPENDIX III)

Governments and interested international organizations are invited to comment on the above standard and should do so in writing in conformity with the Guide to the Consideration of Standards at Step 8 (See *Procedural Manual of the Codex Alimentarius Commission*, Eleventh Edition, page 26-27) to the Secretary of the Codex Alimentarius Commission, Joint FAO/WHO Food Standards Programme, FAO, Viale delle Terme di Caracalla,00100 Rome, Italy, Fax: +39 06570 54593, Email: Codex@fao.org <u>before 1st February 2001</u>

3. DRAFT STANDARD FOR COCOA POWDERS (COCOAS) AND DRY MIXTURES OF COCOA AND SUGARS AT STEP 8 (ALINORM 01/14 - PARAS 28-46 AND APPENDIX IV)

Governments and interested international organizations are invited to comment on the above standard and should do so in writing in conformity with the Guide to the Consideration of Standards at Step 8 (See *Procedural Manual of the Codex Alimentarius Commission*, Eleventh Edition, page 26-27) to the Secretary of the Codex Alimentarius Commission, Joint FAO/WHO Food Standards Programme, FAO, Viale delle Terme di Caracalla,00100 Rome, Italy, Fax: +39 06570 54593, Email: Codex@fao.org before 1st February 2001

PART B REQUEST FOR INFORMATION AND COMMENTS

1. PROPOSED DRAFT STANDARD FOR CHOCOLATE AND CHOCOLATE PRODUCTS AT STEP 5 (ALINORM 01/14 - PARAS 47-75 AND APPENDIX V)

Governments and interested international organizations wishing to submit comments on the above standard at Step 5 should do so in writing in conformity with the Uniform Procedure for the Elaboration of Codex Standards and Related Texts (See *Procedural Manual of the Codex Alimentarius Commission*, Eleventh Edition, page 23-24) to Ms Awilo Ochieng Pernet ; Codex Contact Point ; Swiss Federal Office of Public Health ; CH-3003 Bern, Switzerland, Fax: +41 31 322 95 74 ; Email: awilo.ochieng@bag.admin.ch with copy to the Secretary of the Codex Alimentarius Commission, Joint FAO/WHO Food Standards Programme, FAO, Viale delle Terme di Caracalla,00100 Rome, Italy, Fax: +39 06570 54593, Email: Codex@fao.org before 1st February 2001

SUMMARY AND CONCLUSIONS

The Eighteenth Session of the Codex Committee on Cocoa Products and Chocolate reached the following conclusions:

The Codex Committee agreed to:

Part A: MATTERS FOR CONSIDERATION BY THE COMMISSION

- (a) Advance the revised *Draft Standard for Cocoa Butter* to Step 8 of the Codex Procedure for final adoption by the Commission. (Para. 16 and Appendix II);
- (b) Advance the revised *Draft Standard for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake* to Step 8 of the Codex Procedure for final adoption by the Commission. (Para. 27 and Appendix III);
- (c) Advance the revised *Draft Standard for Cocoa Powders (Cocoas) and Dry Mixtures* of *Cocoa and Sugars* to Step 8 of the Codex Procedure for final adoption by the Commission (Para. 46 and Appendix IV)
- (d) Ask the Commission to revoke all the existing standards on Chocolate and Chocolate products for which that are covered by the Draft Standards above proposed for adoption.
- (e) Advance the revised Proposed Draft Standard for Chocolate and Chocolate Products for consideration by the Commission at Step 5 of the Codex Procedure (Para. 75 and Appendix V), noting that considerable progress had been made on the most difficult issues surrounding this standard.

Part B: MATTERS OF INTEREST TO THE CODEX COMMITTEE ON FOOD ADDITIVES AND CONTAMINANTS

- (f) Invite Member Countries to provide data on Lead levels in products obtained from good agricultural and manufacturing practices, and to submit these data to CCFAC to either confirm the levels of Lead or to justify higher levels established in Cocoa Products and Chocolate (See paras. 11-13; 24 and 40);
- (g) Seek the advice of the CCFAC on a list of food additives intended to be used in the preparation of Cocoa Powders (Cocoas) and Dry Mixtures of Cocoa and Sugarsbased Beverages. (See para. 35)
- (h) Submit complete lists of food additives intended to be endorsed in the preparation of Cocoa Products and Chocolate in the above standards.

-vi-

TABLE OF CONTENT

INTRODUCTION1
ADOPTION OF THE AGENDA (AGENDA ITEM 1)1
MATTERS REFERRED TO THE COMMITTEE BY THE CODEX ALIMENTARIUS COMMISSION AND/OR OTHER CODEX COMMITTEES (AGENDA ITEM 2)1
CONSIDERATION OF DRAFT REVISED STANDARDS AT STEP 7 (AGENDA ITEM 3)2
CONSIDERATION OF THE PROPOSED DRAFT REVISED STANDARD FOR CHOCOLATE AND CHOCOLATE PRODUCTS AT STEP 4 (A GENDA I TEM 4)
OTHER BUSINESS AND FUTURE WORK (AGENDA ITEM 5)11
DATE AND PLACE OF THE NEXT SESSION (AGENDA ITEM 6)11

LIST OF APPENDICES AND ANNEX

ANNEX I	SUMMARY STATUS OF WORK12
Appendix I	LIST OF PARTICIPANTS
APPENDIX II	DRAFT REVISED STANDARD FOR COCOA BUTTER (AT STEP 8)23
APPENDIX III	DRAFT REVISED STANDARD FOR COCOA (CACAO) MASS COCOA/CHOCOLATE LIQUOR) AND COCOA CAKE (AT STEP 8)
APPENDIX IV	DRAFT REVISED STANDARD FOR COCOA POWDERS (COCOAS) AND DRY MIXTURES OF COCOA AND SUGARS (AT STEP 8)
Appendix V	PROPOSED DRAFT STANDARD FOR CHOCOLATE AND CHOCOLATE PRODUCTS (AT STEP 5)
Appendix VI	LIST OF FOOD ADDITIVES PROPOSED BY THE DELEGATION OF JAPAN FOR USE IN COCOA POWDERS(COCOAS) AND DRY MIXTURES OF COCOA AND SUGARS, AND RELATED PRODUCTS

REPORT OF THE EIGHTEENTH SESSION OF THE CODEX COMMITTEE ON COCOA PRODUCTS AND CHOCOLATE

INTRODUCTION

1. The Codex Committee on Cocoa Products and Chocolate held its Eighteenth Session in Fribourg, Switzerland from 2 to 4 November 2000 at the kind invitation of the Swiss Government. The Session was chaired by Prof. Dr. Ing. Erich J. Windhab, Chair of Food Process Engineering, Institute of Food Science, Swiss Federal Institute of Technology, Zurich. It was attended by 76 Delegates, advisors and observers from 26 Members of the Commission and 12 observers from 6 international governmental and non-governmental organizations. A full list of participants, including members of the Secretariat, is given in Appendix I to this report.

ADOPTION OF THE AGENDA¹ (Item 1)

2. The Committee adopted the Provisional Agenda as the Agenda for the Session.

MATTERS REFERRED TO THE COMMITTEE BY THE CODEX ALIMENTARIUS COMMISSION AND/OR OTHER CODEX COMMITTEES² (ITEM 2)

3. The Committee noted that the 23rd Session of the Commission had adopted the Proposed Draft Standards for Cocoa Butters, for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake for the Use in the Manufacture of Cocoa and Chocolate Products, and for Cocoa Powders (Cocoas) and Dry Cocoa-Sugar Mixture at Step 5 as proposed and advanced them to Step 6. It was noted that these matters would be taken up under Agenda Item 3.

ENDORSEMENT OF FOOD ADDITIVES PROVISIONS

4. The Codex Committee on Food Additives and Contaminants (CCFAC, 31st Session) endorsed the food additives provisions in the Proposed Draft Standards for Cocoa Butters, for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake for the Use in the Manufacture of Cocoa and Chocolate Products, and for Cocoa Powders (Cocoas) and Dry Cocoa-Sugar Mixture, except Potassium aluminium silicate (555) as an anti-caking agent in Cocoa Powders (Cocoas) and Dry Cocoa-Sugar Mixtures since this substance had no Acceptable Daily Intake (ADI) allocated by JECFA.

ENDORSEMENT OF CONTAMINANT PROVISIONS

5. The CCFAC considered the contaminant provisions in the Proposed Draft Standards for Cocoa Butters, for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake for the Use in the Manufacture of Cocoa and Chocolate Products, and for Cocoa Powders (Cocoas) and Dry Cocoa-Sugar Mixture. The CCFAC decided to refer these levels to this Committee for further consideration. Opinions were expressed at the 31st Session of the CCFAC that:

- the maximum levels for lead were too high;
- more information should be provided to the CCFAC to justify the maximum levels of lead; and
- the levels should be based on sound science and reflect what was technologically feasible.

¹ CX/CPC 00/1

² CX/CPC 00/2

STATUS OF CODEX TEXTS IN THE FRAMEWORK OF THE TBT AGREEMENT

6. The Committee noted that this matter had been discussed by the Codex Committee on General Principles (CCGP, 14th Session, 1999) and it had been agreed that all Codex texts, including standards and their annexes were covered by the WTO Agreement on Technical Barriers to Trade.³ It was further noted that under the Agreement, countries were not obliged to implement international standards automatically, but if an international standard existed or their completion was imminent, such standards or relevant parts of them, should be used in the development of any national standards for the same products except where such international standards or relevant parts would be ineffective or inappropriate.⁴

CONSIDERATION OF DRAFT REVISED STANDARDS AT STEP 7⁵ (Agenda Item 3)

A. DRAFT STANDARD FOR COCOA BUTTER (ALINORM 99/14, APPENDIX II)

Title and Scope

7. The Committee adopted the Title and Scope of the standard without change.

Description (Section 2)

8. At the request of the Delegation of India, the Committee decided to reinstate the former maximum level for unsaponifiable matter of 0.7% m/m, in order to meet processing limitations in many countries. The Delegation of France on behalf of the Members of the European Union expressed its reservation at this decision.

9. The Committee decided to recognize the existence in trade of "*press cocoa butter*" as defined in the Standard and agreed to remove the square brackets from the definition of this product.

Food Additives/Processing Aids (Section 3)

10. The Delegation of the United States proposed that all safe and suitable extraction solvents be allowed for use, and this proposal was supported by several other Delegations. It was noted that several extraction solvents were included in the Codex Inventory of Processing Aids with defined maximum limits for residues. The Committee decided however to retain the reference only to hexane with a maximum residue limit of 1 mg/kg as had been endorsed by the CCFAC.

Contaminants (Section 4)

11. The Committee had an extensive debate on the maximum limit for **Lead** (**Pb**) and the possibility of reducing the proposed limit from 0.5 mg/kg to 0.1 mg/kg. Several delegations reported that levels found in commerce indicated that the level of 0.1 mg/kg could be achieved with good agricultural and manufacturing practices. Other Delegations stated that levels of less than 0.5 mg/kg were not technically achievable in developing countries in particular, and that levels ranging as high a 1 mg/kg were sometimes encountered. It was noted that the Draft Maximum

³ ALINORM 99/33A, para. 61.

⁴ Agreement on Technical Barriers to Trade, Article 2.4.

⁵ ALINORM 99/14, Appendices I, II and III; Circular Letter CL 1999/13-CPC; CX/CPC 00/3 (Comments of Cuba, Switzerland, Thailand, Netherlands, CAOBISCO); CX/CPC 00/3-Add.1 (Comments of Japan, European Community); CRD 1 (Comments of India); CRD 2 (Comments of Canada); CRD 3 (Comments of Hungary); CRD 5 (Comments of The Philippines).

Limits for Lead in Food⁶ that had been submitted to the Commission for adoption included the limit of 0.1 mg/kg for vegetable fats.

12. In proposing this level, the Committee invited countries that had data on Lead levels in products obtained from good agricultural and manufacturing practices, to submit these data to the CCFAC to either confirm the level of 0.1 mg/kg, or to justify a higher level.

13. The Committee noted the comment of the Delegation of India that the reduction of the limit would have to be foreseen within a reasonable timeframe to allow for changes to agricultural, processing and production practices.

14. The Committee did not adopt a proposal from Switzerland to include a limit of 20 mg/kg for aliphatic and aromatic hydrocarbons originating from jute bags. Although recognizing the problem of the occurrence of these substances in cocoa products, the Committee also recognized that changes in the mode of packaging and transport that avoided the use of jute bags were coming into effect and that as a consequence the problem would diminish. The Committee agreed that it would be premature to include maximum limits, but encouraged countries to take further measures to reduce the presence of these substances in cocoa products.

Labelling (Section 6)

15. The Committee agreed to provide for the labelling of both "*cocoa butter*" and "*press cocoa butter*" according to the criteria specified in Section 2 of the Standard.

Status of the Proposed Draft Standard for Cocoa Butter

16. The Committee agreed to advance the revised Draft Standard for Cocoa Butter to Step 8 of the Codex Procedure for final adoption by the Commission. The full text of the revised Draft Standard is contained in Appendix II of this report.

B. DRAFT STANDARD FOR COCOA, COCOA (CACAO) MASS (COCOA/CHOCOLATE LIQUOR) AND COCOA CAKE FOR USE IN THE MANUFACTURE OF COCOA AND CHOCOLATE PRODUCTS (ALINORM 99/14, APPENDIX III)

Title and Scope

17. The Committee noted the opinions of those Delegations that wished to add synonyms for cocoa mass to the title of the standard to make reference specifically to *cocoa paste, unsweetened chocolate* and *bitter chocolate*. The Committee amended the Title and Scope of the standard to deal with the consequences of the introduction of the product *unsweetened chocolate* in the detailed sections of the standard dealing with composition and labelling.

Descriptions (Section 2)

18. The Committee agreed to include a reference to the possibility of roasting and the possibility of adding or removing constituents of the cocoa nib. The Delegation of Spain expressed its disagreement with the possibility of removal of cocoa constituents as this would allow for the removal of cocoa butter.

Essential Composition and Quality Factors (Section 3)

19. The Committee decided to separate the requirements for Cocoa Mass and Cocoa Cake in order to provide limits for the cocoa butter content of Cocoa Mass (47 to 60% m/m). The

6

ALINORM 01/12, Appendix XI "Draft Maximum Limits for Lead at Step 8".

Committee agreed to remove the requirements for *Total Ash* and *Ash Insoluble in Hydrochloric Acid* for both Cocoa Mass and Cocoa Cake. The Delegation of The Netherlands noted that this would remove a useful control over the presence on mineral impurities in the products concerned.

Food Additives (Section 4)

20. The Committee had an extended discussion on the relationship between the food additives provisions in commodity standards and those in the General Standard for Food Additives (GSFA). It was noted that the GSFA was a "work in progress" and that many specific food and food additive combinations had yet to be finalized. For the time being, the food additive provisions in Codex Standards were being included in the GSFA as the appropriate provisions for the commodities covered by Codex commodity standards. When the GSFA was closer to completion, it would be necessary to rationalize the general provisions of the GSFA with specific provisions for commodity standards, taking into account the technological needs for the food additives.

21. The Delegation of Germany expressed its reservation concerning the use of artificial flavours, and the Delegation of Italy expressed the opinion that it would like to include artificial flavours in a separate section.

22. The Committee agreed to add Magnesium Oxide (INS 530) to the list of Acidity Regulators and Polyglycerol Esters of Interesterified Ricinoleic Acid (INS 476) to the list of Emulsifiers.

23. The Observer from the European Community stated that as the products covered by the standard were intermediate products for further manufacturing, there should either be no need to specify maximum limits or that the maximum limits should be those expressed on the finished product basis. The Committee agreed that the numeric levels given should represent the level in the finished cocoa or chocolate product.

Contaminants (Section 5)

24. The Committee, noting its discussion in relation to Cocoa Butter (see paras. 11-13 above), decided to reduce the level for Lead to 1 mg/kg in these products and invited countries that had data on Lead levels, to submit these data to the CCFAC.

Labelling (Section 7)

25. The Committee agreed to include a reference to the use of the terms *Cocoa Paste*, *Unsweetened Chocolate*, and *Bitter Chocolate* as synonyms for Cocoa (Cacao) Mass and Cocoa/Chocolate Liquor provided that this was acceptable under the requirements of the legislation in the importing country.

Methods of Analysis and Sampling

26. The Committee agreed to add a method for the determination of Fat Content and deleted the methods for the determination of Total Ash and Ash Insoluble in Hydrochloric Acid.

Status of the Proposed Draft Standard for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake

27. The Committee agreed to advance the revised Draft Standard for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake to Step 8 of the Codex Procedure for final adoption by the Commission. The full text of the revised Draft Standard is contained in Appendix III of this report.

C. DRAFT STANDARD FOR COCOA POWDERS (COCOAS) AND DRY COCOA-SUGAR MIXTURES (ALINORM 99/14, APPENDIX IV)

Title and Scope

28. The Committee amended the Title and Scope to indicate that the products were dry mixtures of cocoa and sugars, not only sucrose.

Essential Composition and Quality Factors (Section 3)

29. The Committee had an extended debate on the Moisture Content of the products covered by the Standard. Several Delegations proposed raising the level of moisture content to 9% m/m to take into account the absorption of moisture during transport, storage and handling before the point of sale to the consumer. Other Delegations proposed a reduction of the moisture content to limit the potential for microbial or mould growth, but data were not available to indicate what level would be required for this purpose. The Committee agreed to retain the current proposed limit of 7% m/m and removed the square brackets surrounding this provision.

30. In order to reflect the Committee's decision concerning the use of sugars other than sucrose, it was agreed to use the term Dry Mixtures of Cocoa and Sugars in place of Cocoa-Sugar Mixtures (Section 3.1.2).

31. The Committee also discussed the differentiation of the products covered by the standard by their cocoa butter content, with some Delegations indicating that the term "*fat-reduced*" applied to products of less than 20% cocoa butter, and other indicating that the term applied to products of less than 10% cocoa butter. Some concerns were expressed about the use of descriptors that might be misleading to consumers. In order to avoid a situation where the same product could be marketed under two competing designations, the Committee decided to establish a three-tier structure as follows:

- Cocoa powder (and synonyms): more than 20% m/m cocoa butter;
- Fat-reduced cocoa powder (and synonyms): between 10% and 20% m/m cocoa butter;
- Highly fat-reduced cocoa powder (and synonyms): less than 10% cocoa butter.

32. It was noted that the term "Drinking Chocolate" ("Chocolat de ménage en poudre") was an acceptable synonym for Sweetened Cocoa Powder in both the full-fat and fat-reduced forms.

33. The Committee decided to present all possible combinations of products depending on their cocoa butter and cocoa powder contents in a tabular form.

34. The Committee decided to retain the definition of the product "*Chocolate Powder*", but amended the minimum cocoa powder content to 32% m/m on the whole product, or 29% on a dry matter basis.

Food Additives (Section 4)

35. The Delegation of Japan submitted a list of food additives in writing and it was recommended to send data on technological justification for the use of such additives to the Codex Committee on Food Additives and Contaminants. The Japan's list is included as Annex VI to this report.

36. The Committee noted the reservation of the Delegation of Germany concerning the use of artificial flavours in these products.

37. The Committee agreed to delete the Anticaking Agent Potassium Aluminium Silicate (INS 555) as recommended by the CCFAC.

- 38. The Committee agreed to add the following substances to the list of food additives:
 - Acidity Regulators: Magnesium Oxide (INS 350) and Calcium Hydroxide (INS 526)
 - Emulsifiers: Polyglycerol Esters of Interesterified Ricinoleic Acid (INS 476)
 - Bulking Agents: Polydextroses A and N (INS 1200)
 - Sweeteners: Acesulfame Potassium (INS 950), Aspartame (INS 951), Sucralose (INS 955), Sodium Saccharin (INS 954), Thaumatin (INS 957)

39. The Committee noted that the inclusion of high-intensity sweeteners in the list of food additives would require additional labelling provisions to ensure that consumers were fully informed of the nature of the products concerned (see below, para. 42). The Delegation of Canada noted that Sodium Saccharin was not allowed for use in that country as a food additive.

40. The Committee agreed that all numeric values given in the list of food additives should refer to the concentration in the finished product. It was further noted that the revised list of food additives would need to be submitted to the CCFAC for endorsement of the changes made to the list.

Contaminants (Section 5)

41. The Committee, noting its discussion in relation to Cocoa Butter (see paras. 11-13 above), decided to reduce the level for Lead to 1 mg/kg in these products and invited countries that had data on Lead levels, to submit these data to the CCFAC.

Labelling (Section 7)

42. The Committee agreed to simplify the provisions concerning the Name of the Food for Dry Mixtures of Cocoa and Sugars by making direct reference to the names given in the table of descriptions in Section 3.1.2.

43. In light of its decision to allow the use of artificial sweeteners, the Committee agreed to include two statements under the Section on Labelling; the first to require that when artificial sweeteners were used to replace all or part of the sugars component, the word *Sugar* or its equivalent should not form part of the Name of the Food but should be replaced in those languages which required it, by the term "*Sweetened*". A second statement was included to require that the term "artificially sweetened" or an equivalent declaration be used in association with the Name of the Food.

44. The Committee agreed that the declaration of minimum cocoa powder content (whether fullfat, fat-reduced, or highly fat-reduced) be declared on the label in accordance with Section 5.1.1 of the General Standard for the Labelling of Prepackaged Foods.

Methods of Analysis and Sampling (Section 8)

45. The Committee agreed to up-date the references to methods for Determination of Cocoa Butter and Determination of Moisture and to delete reference to methods for which there were no criteria established in the standard. It was noted that there was no current method for the determination of cocoa powder (whether full-fat, fat-reduced, or highly fat-reduced), but the Committee decided to retain the possibility of the development of such a method in the future to enable officials to determine compliance with the provisions of the standard.

Status of the Proposed Draft Standard for Cocoa Powders (Cocoas) and Dry Mixtures of Cocoa and Sugars

46. The Committee agreed to advance the revised Draft Standard for Cocoa Powders (Cocoas) and Dry Mixtures of Cocoa and Sugars to Step 8 of the Codex Procedure for final adoption by the Commission. The full text of the revised Draft Standard is contained in Appendix IV of this report.

CONSIDERATION OF THE PROPOSED DRAFT REVISED STANDARD FOR CHOCOLATE AND CHOCOLATE PRODUCTS AT STEP 4⁷ (Agenda Item 4)

TITLE AND SCOPE

47. The Committee agreed to simplify the Scope of the Standard based on a proposal provided by the Delegation of Malaysia in its written comments.

DESCRIPTION AND ESSENTIAL COMPOSITION AND QUALITY FACTORS (SECTION 2)

2.1 Chocolate Types

48. The Committee discussed the nature of the edible foodstuffs that may be added to chocolate to form chocolate products. It was noted that the foodstuffs could be in the nature of particulate products (e.g., fruit pieces or nuts) or homogeneous products such as nut pastes. Several suggestions were made to clarify this in the Description, but it was agreed to retain the current wording of the paragraph.

Vegetable fats other than cocoa butter

49. The Committee discussed the proposal to admit the use of up to 5% of vegetable fats other than cocoa butter in the finished product under certain conditions. The Delegation of France, speaking on behalf of the Members of the European Union present at the Session, supported the proposal which it indicated was consistent with recent changes in European legislation, but indicated that the inclusion of such a provision was conditional on the introduction of suitable labelling provisions and an assurance that only suitable vegetable fats would be used. This position was supported by a number of Delegations, including those of Benin and Burkina Faso. These Delegations noted that the proposal as presented fulfilled the conditions set out by the Delegation of Côte d'Ivoire at the Committee's 17th Session⁸.

50. Several other Delegations, including those of Canada, India and the USA, indicated that national legislation did not, at present, allow the use of vegetable fats other than cocoa butter, but that they could support the proposal in the interest of international trade. The Delegation of Canada stated however that there was no consensus on this issue by the Chocolate industry in Canada. Some other Delegations, including Japan, stated that there should be no quantitative restriction to the use of edible vegetable fats and oils in the products covered by the standard in order to incorporate modern technology, to create innovative Chocolate products, and to prevent melting and fat bloom in Chocolate products.

51. The Delegation of Cameroon, supported by the representative of COPAL, stated that the proposal represented a negative social and economic step for cocoa-producing countries. They

⁷ ALINORM 99/14, Appendix V; CL 1998/43-CPC; CX/CPC 00/4 (Comments of Brazil, Cuba, Japan, Mexico, Spain, Thailand, United Kingdom, CAOBISCO); CX/CPC 00/4-Add.1 (Comments of Italy, Republic of Korea, Malaysia, CAOBISCO); CX/CPC 00/4-Add.2 (Comments of Japan, European Community); CRD 1 (Comments of India); CRD 2 (Comments of Canada); CRD 3 (Comments of Hungary); CRD 4 (Comments of The Philippines).

⁸ ALINORM 99/14, para. 12.

noted that due to unfulfilled market expectations in importing countries there was considerable over-production of cocoa beans and that the application of the present proposal would aggravate the situation. They stated that the addition of vegetable fats would give rise to a range of products that could no longer be considered as "chocolate" in the sense of the traditional use of the term.

52. The Representative of COPAL also raised the question of the availability of suitable methods of analysis to ensure compliance with the proposed 5% limit. The Delegation of the United Kingdom stated that such methods were available and had been shown to be precise and reliable when used in combination with normal food inspection procedures. The Delegation claimed that these methods were accurate to $\pm 0.3\%$. Copies of published methods were provided to delegates for information.⁹ The Representative of COPAL reserved the opinion of that organization pending review of the methods.

53. The Committee decided to remove the square brackets from the proposal to allow the use of up to 5% vegetable fats other than cocoa butter. The Representative of COPAL expressed the strong reservation of that organization.

54. The Committee had an extended discussion on that nature of the vegetables fats that could be used on chocolate and chocolate products as described above. Several Delegations and the Representatives of the EC and COPAL stressed that the fats used should be technically compatible with cocoa butter, should be from specified sources processed under specific conditions. Other Delegations stated that manufacturers should have the flexibility of choice to enable them to meet differing market requirements and climatic conditions. The Committee agreed to a compromise, proposed by Canada, to leave the specific rules governing the choice of vegetable fats to national legislation, thereby meeting local traditions and cultural expectations.

55. Finally the Committee discussed the labelling that would be required in order implement the introduction of the use of vegetable fats. Although some Delegations were of the opinion that listing in the List of Ingredients would normally be sufficient, the Committee recognized that the introduction of the use of 5% of vegetable fat was based on a more general compromise that included the need for labelling. The Committee agreed to a compromise text that required the addition of vegetable fats to be indicated clearly on the label in association with the name or representation of the food, but leaving to national authorities the precise manner of the declaration.

Specific Chocolate Types

56. It was noted that the product *Unsweetened Chocolate* had been removed from the standard and included in the Draft Standard for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake

(see para. 17 above). All references to this product were consequently deleted from the text.

57. Several Delegations, including Japan, proposed a simplification of the standard to provide only for the major general types of products moving in international trade, namely Chocolate, Milk Chocolate and Cocoa Butter Confectionery (White Chocolate). It was also suggested that this list should be extended to cover Filled Chocolate. The Committee examined these proposals in detail and considered which among the individual products named in the standard should be retained and which should be deleted. The Committee finally agreed that all products of importance to trade for

These were:

9

Macarthur, R., C. Crews and P. Brereton An improved method for the measurement of added vegetable fats in chocolate. Food Additives and Contaminants Vol 17, No. 8, pp 643-664 (2000); and

Lipp, M. and E. Anklam: Review of cocoa butter and alternative fats for use in chocolate – Part B. Analytical approaches for identification and determination. Food Chemistry, Vol.62 No.1 pp 99-108 (1998).

Member countries should be included specifically. Specific provisions Skimmed Milk Chocolate, Cream Chocolate and Cocoa Butter Confectionary Vermicelli were deleted.

Sweet Chocolate

58. The Committee deleted the alternative designation "Plain" in order to avoid confusion with other chocolates types.

Milk Chocolate

59. The Committee considered the meaning of the term *milk solids* and agreed that the minimum addition of milk solids referred to the addition of milk ingredients in their natural proportions except that milk fat may be added or removed. Nevertheless, the Committee could not agree on the minimum addition to be specified in the standard, with some Delegations proposing 12% and some 14%. Similarly, some Delegations proposed a higher minimum of milk fat (3.5%), while others proposed retaining the lower level contained in the proposed draft (2.5%). The Committee agreed to a formulation that would allow the setting of minimum values by national authorities within the ranges 12% to 14 % for milk solids, and 2.5% to 3.5% for milk fat.

60. The Committee agreed that Milk Chocolate with a High Milk Content should be included in the standard under the name "Family Milk Chocolate" (Chocolat de Ménage au lait).

61. A new definition for Milk Couverture Chocolate was included.

Cocoa Butter Confectionary/White Chocolate

62. The Committee agreed to a formulation that would allow the setting of minimum values for milk fat by national authorities within the range 2.5% to 3.5%.

Vermicelli and Flakes

63. The Committee noted the need for designations for certain products of significance in regional trade and proposed by the Delegations of Brazil, Spain and Venezuela and included the names of these products as alternatives.

Filled Chocolate

64. Several Delegations, including Japan and India, raised the problem of the correct interpretation of the sentence in the Definition that states: "Filled Chocolate does not include Flour Confectionary, Pastry, Biscuit or Ice Cream Products" and suggested its deletion. Other Delegations pointed out that this exclusion was required to avoid confusion with Filled Chocolate and other products covered or enrobed with chocolate but which did not meet the essential composition requirements. It was pointed out that such products could include the use of the word "chocolate" in their name under the conditions of Section 6.1.8 of the standard. On this basis, it was agreed to retain the sentence.

Other Chocolate Products

65. The Committee agreed to the use of the term "Praline" as a synonym for "A Chocolate". The corresponding term in French was identified as "Bonbon de chocolat/Praline".

66. In reference to the definition of *Gianduja Chocolate*, the Committee noted a proposal to include a maximum limit for the addition of hazelnut to 40% of the total product. It was noted however that this was a requirement established in Section 2.1 and which covered all products; a specific reference in the definition for Gianduja Chocolate was therefore considered redundant.

FOOD ADDITIVES (SECTION 3)

67. The Committee agreed to include in this Section all of the Food Additives approved and endorsed in the standards for Cocoa Butter and for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake, as these were raw material in the production of chocolate and chocolate products.

68. The Committee agreed to delete reference to certain sweeteners that had not as yet been evaluated by JECFA. On the other hand, it agreed to retain reference to the use of metallic Gold (INS 175) and Silver (INS 174) as it noted that these had been evaluated by JECFA in 1977 and were not considered to represent a health hazard. It also decided to delete the square brackets surrounding the addition of Antioxidants to Cocoa Butter Confectionary/White Chocolate (Section 3.6). The Delegation of France on behalf of the Members of the European Union and the Representative of the EC expressed a reservation on this matter.

69. On this basis the Committee agreed to forward the list of food additives to the CCFAC for endorsement. It noted that supplementary lists of additives, submitted by Japan in particular in written comments, would have to be considered in detail at the Committee's next session.

CONTAMINANTS (SECTION 4)

70. The Committee agreed to delete the contaminant provisions for Unsweetened Chocolate as a consequence of the removal of this product from the standard.

LABELLING (SECTION 6)

Name of the Food

71. The Committee agreed to include the following names in square brackets as synonyms for Chocolate as defined in Section 2.1.1: *'bittersweet chocolate*'', *'semi-sweet chocolate*'', and *'dark chocolate*'', to accommodate the common use of these names in some countries. The Delegations of Germany and the United Kingdom disagreed with this proposal.

72. The Delegation of Italy proposed the deletion of the first part of Section 6.1.2 in order not to complicate the common names of foods and to delete Section 6.1.8 entirely. The Committee agreed to discuss these matters at its next session.

73. As noted above (see para. 54), the Committee included specific wording covering the labelling of products to which vegetable fats other than cocoa butter had been added.

METHODS OF ANALYSIS AND SAMPLING (SECTION 7)

74. The Committee agreed to include a reference to a suitable method for the detection/determination of vegetable fats, with the method yet to be identified, and deleted references to methods for the determination of sugars and total ash.

STATUS OF THE PROPOSED DRAFT STANDARD FOR CHOCOLATE AND CHOCOLATE PRODUCTS

75. The Committee, noting that considerable progress had been made on the most difficult issues surrounding the standard, agreed to advance it for consideration by the Commission at Step 5 of the Codex Procedure. The full text of the Proposed Draft Standard is given in Appendix V of this report.

OTHER BUSINESS AND FUTURE WORK (Agenda Item 5)

76. There was no other business. The Committee noted that its future work would consist of the finalization of the Proposed Draft Standard for Chocolate and Chocolate Products.

DATE AND PLACE OF THE NEXT SESSION (Agenda Item 6)

77. The Committee noted that its next Session would be held in Switzerland at a date and place to be agreed upon by the Swiss and Codex Secretariats.

ANNEX I

SUMMARY STATUS OF WORK

Subject	Step	Action by	Reference in ALINORM 01/20
Draft Standard for Cocoa Butter	8	CAC 24 th Session	Para. 16 and Appendix II
Draft Standard for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake	8	CAC 24 th Session	Para. 27 and Appendix III
Draft Standard for Cocoa Powders (Cocoas) and Dry Mixtures of Cocoa and Sugars	8	CAC 24 th Session	Para. 46 and Appendix IV
Proposed Draft Standard for Chocolate and Chocolate Products	5	CAC 24 th Session CPC 19 TH Session	Para. 75 and Appendix V
Lead levels in Standards	For endorsement	CCFAC	Paras. 11-13 ; 24 and 40
Food additives lists	For endorsement	CCFAC	Paras. 10, 20, 40 and 69 and Appendix VI

APPENDIX I

LIST OF PARTICIPANTS LISTE DES PARTICIPANTS LISTA DE PARTICIPANTES

CHAIRPERSON/PRESIDENT/PRESIDENTE

Prof. Dr. Ing. Erich J. Windhab Swiss Federal Institute of Technology Zurich Institute of Food Science ETH-Zentrum, LFO E 18 Rämistrasse 10 CH-8029 Zürich **Phone:** +41 1 632 53 48 **Fax:** +41 1 632 11 55 **Email:** windhab@ilw.agrl.ethz.ch

ASSISTANT TO THE CHAIRPERSON/ASSISTANTE DU PRESIDENT/ASSISTENTE AL PRESIDENTE

Eva Zbinden Kaessner, Attorney Head International Standards Unit Swiss Federal Office of Public Health 3003 Berne Phone: +41 31 322 95 72 Fax: +41 31 322 95 74 Email: eva.zbinden@bag.admin.ch

SECRETARIAT/SECRETARIAT/SECRETARIÀ

Awilo Ochieng Pernet, lic. In law Codex Alimentarius International Standards Unit Swiss Federal Office of Public Health 3003 Berne Phone: +41 31 322 00 41 Fax: +41 31 322 95 74 Email: awilo.ochieng@bag.admin.ch

Heads of Delegation are listed firster.

Les chefs de délégation figurent en tête.

Figuran en primer lugar los Jefes de las delegaciones.

page 14

MEMBER COUNTRIES PAYS MEMBRES PAISES MEMBROS

AUSTRALIA AUSTRALIE

Dr. Simon Brooke-Taylor Program Manager Product Standards Australia New Zealand Food Authority (ANZFA) PO Box 7186 Canberra Mail Centre ACT 2610 **Phone:** 61 2 6271 2225 **Fax:** 61 2 6271 2278 **Email:** simon.brooke-taylor@anzfa.gov.au

AUSTRIA

AUTRICHE

Dr. Gertraud Fischinger Permanent Representation of Austria to the EU Avenue de Cortenberg 30 B-1040 Brussels Phone: +32 2 2345 221 Fax: +32 2 2345 311 Email: gertraud.fischinger@bmaa.gv.at

BENIN Bénin Benin

M. Aristide Sagbohan Medecin Nutritionniste Dirécteur de l'Alimentation et de la Nutrition Appliquée Ministère du Développement Rural BP 295 Porto-Novo **Phone:** +229 21 26 70/33 01 23 **Fax:** +229 31 39 63

M. Yessoufou Alamon Chef du Service Qualité, Analyses et Législation Alimentiares Ministère du Développement Rural Direction de l'Alimentation et de la Nutrition Appliquée BP 295 Porto-Novo Phone: +229 21 26 70/22 58 01 Fax: +229 21 39 63 Email: danamdr@leland.bj

BRAZIL BRESIL BRASIL

M. Paulo Traballi-Bozzi Counsellor Brazilian Embassy in Bern Monbijoustrasse 68 CH-3007 Bern, Switzerland Phone: +41 31 371 85 15 Fax: +41 31 371 05 25 Email: brasbern@iprolink.ch

BURKINA FASO

M. Ambroise Balima Conseiller Economique Ambassade du Burkina Faso Place Guy d'Arezzo 16 B- 1180 Bruxelles, Belgique Phone: +32 2 345 99 12 Fax: +32 2 345 06 12 Email: ambassade.burkina@skynet.be

CAMEROON CAMEROUN CAMERÚN

M. Goldfine Mathié Bomia Directeur Général par intérim de l' Office National du cacao et du Café

B.O. Box 3018 Douala

 Phone:
 +237 43 1978/42 6776

 Fax:
 +237 43 42 0002/1596

 Email
 bomia_2000@yahoo.com

CANADA CANADÁ

Ms Claudette Dalpé Head of Delegation Associate Director Food Regulatory Program Bureau of Food Regulatory, International and Interagency Affairs Food Directorate Health Products and Food Branch, Health Canada Ottawa, Ontario K1A 0L2 Locator (0702C1) Phone: +1 613 957 1750 Fax: +1 613 941 3537 Email: claudette_dalpe@hc-sc.gc.ca

Mr Laurent Laflamme Confectionery Manufacturers Association of Canada Technical Committee Chairman 885 Don Mills Road, Str. 301 Don Mills, Ontario M3C IV9 Phone: +1 416 510 8034 Email: laurent_laflamme@barrycallebaut.com

DENMARK Danemark Dinamarca

Ms Marianne Treumer Andersen Legal Adviser Ministry of Food, Agriculture and Fisheries Danish Veterinary and Food Administration Moerkhoej Bygade 19 DK-2860 Soeborg Phone: +45 33 95 60 00 Fax: +45 33 95 60 01 Email: mta@fdir.dk

FRANCE FRANCIA

M. Jean-Jacques Ambroise Chef de bureau Direction Générale Concurrence, Consommation, Répression des Fraudes (DGCCRF) 59, boulevard Vincent Auriol 75013 Paris Phone: +33 1 44 97 23 23 Fax: +33 1 44 97 05 27 Email: jeanjacques.ambroise@dgccrf.finances.gouv.fr

Dr. Carol Buy Chargée de mission Ministère de l'Agriculture et de la Pêche Direction Générale de l'Alimentation 251, rue de Vaugirard 75732 Paris Cedex **Phone:** +33 1 49 55 58 63 **Fax:** +33 1 49 55 59 48 **Email:** carol.buy@agriculture.gouv.fr

Mme Roseline Lecourt Ministère de l'économie, des finances et de l'industrie 59 Bld Vincent Auriol 75703 Paris Cedex 13 Phone: +33 1 44 97 34 70 Fax: +33 1 44 97 30 37 Email: lecourt.roseline@dgccrf.finances.gouv.fr

Mme Catherine Chapalain Chef du service réglementation Chambre syndicale des chocolatiers/Alliance 7 194, rue de Rivoli 75001 Paris Phone: +33 1 44 77 85 85 Fax: + Email: cchapalain@alliance7.com M. Phi Phung NGuyen Chargé de mission Ministère de l'Agriculture et de la Pêche Direction des politiques économique et internationale 3, rue Barbet de Jouy 75007 Paris Cedex **Phone:** +33 1 49 55 58 60 +33 1 49 55 50 75 Fax: Email: phi-phung.nguyen@agriculture.gouv.fr. Mme Liane Ude Product Design + Legal Manager Mars Alimentaire S.A. 3. chemin de la Sandlach 67500 Haguenau **Phone:** +33 3 88 05 13 12 +33 3 88 05 10 06 Fax: Email: liane.ude@eu.effem.com Mme Hilde Van Gerwen International Legislation Manager Barry Callebaut France 5, boulevard Michelet 78250 Meulan **Phone:** +33 1 30 22 84 42 Fax: +33 1 30 22 87 56 Email: hilde_van_gerwen@barrycallebaut.com GERMANY ALLEMAGNE ALEMANIA Mr.Hermann Brei Regierungsdirektor Bundesministerium für Gesundheit Am Propsthof 78a D-53121 Bonn **Phone:** +49 228 941-4141 Fax: +49 228 941-4842 Email: brei@bmg.bund.de Mr Hagen Streichert Oberregierungsrat Bundesministerium für Ernährung, Landwirtschaft und Forsten Rochusstrasse 1 D-53123 Bonn **Phone:** +49 228 529 36 48 Fax: +49 228 529 44 10 Email: hagen.streichert@bml.bund.de Dr. Dietmar Kendziur Geschäftsführender Direktor Ferrero Deutschland Hainerweg 120 D-60599 Frankfurt am Main **Phone:** +49 69 680 5274 Fax: +49 69 680 5353

Dr. Karsten Keunecke Geschäftsführer Bundesverband der Deutschen Süsswarenindustrie e.V. Schumannstrasse 4 - 6 D-53113 Bonn Phone: +49 228 2600722 Fax: +49 228 2600789 Email: karsten.keunecke@bdsi.de Mr Gerhard Schweer-Lambers

Bundesverband der Deutschen Süsswarenindustrie e.V. Schumannstrasse 4 - 6 D-53113 Bonn Phone: +49 228 2600770 Fax: +49 228 2600789

HUNGARY HONGRIE HUNGRÍA

Dr. Mohos Ferenc Expert Ministry of Agriculture and Regional Development Kossuth Tèr 11. H-1055 Budapest Phone: +36 1 375 9911/ +36 309 324 287 Fax: +36 1 375 9911 Email: edesseg@matavnet.hu

INDIA INDE

Mr Kumar Tuhin First Secretary Permanent Mission of India to the United Nations Offices 9, rue du Valais 1202 Genève Phone: + 41 22 906 86 86 Fax: + 41 22 906 86 96

Mr Ajit Tawde General Technical Manager Cadbury India Limited 19, B. Desai Road Mahalaxmi Mumbai Phone: +00 91 22 4939 558 Fax: +00 91 22 4938 466 Email: ajit.tawde@csplc.com

ITALY ITALIE ITALIA

Dr Giuseppe De Giovanni Director Ministero Industria Via Molise 2 00187 Roma Phone: +39 06 47887729 Fax: +39 06 47887797 Drssa Brunella Lo Turco Segretario Generale Comitato Nazionale Codex Ministero delle Politiche Agricole Via Sallustiana 10 00187 Rome Phone: +39 06 488 02 73 Fax: +39 06 488 02 73 Email: blturco@tiscalinet.it Dr. Luca Ragaglini Esperto tecnico legale Associazione Industrie Dolciarie Italiane AIDI Via Barnaba Oriani 92 Rome **Phone:** +39 06 809 10 71 Fax: +39 06 807 31 86 Email: aidi@foodarea.it Dr. Alix Lauer Corporate Counsel Soremartec, Groupe Ferrero 187, ch. de la Hulpe B- 1170 Bruxelles, Belgium **Phone:** +32 2 679 04 24 Fax: **Email**: alix.lauer@skynet.be JAPAN **JAPON** JAPÓN Ms Noriko Matsunami Planning Section Chief Groceries, Oils & Fats Division Japanese Ministry of agriculture, Forestry and Fisheries 1-2-1 Kasumigaseki Chiyoda-ku Tokyo 100-8950 **Phone:** +81 3 3501 3815 Fax: +81 3 3502 0614 Email: noriko.matsunami@nm.maff.go.jp Mr Masanori Ito **Technical Advisor** National Association of Food Industry Lotte Co. Ltd., Central Laboratory 3-1-1, Numakage, urawa, 336-0027 Saitama **Phone:** +81 48 837 0247 Fax: +81 48 837 0130 Mr Kazuo Kishida Technical Advisor National Association of Food Industry 4-16 Kyobashi 2-Chome, Chuo-Ku Tokyo 104-8802 Phone: +81 3 3273 3381 Fax: +81 3 3279 5736 Email: kazuo-kishida@meiji.co.jp

Mr Yoichi Tashiro National Association of Food Industry 2-3-45, Takamtsu, Izumi Sano Osaka-Fu **Phone:** +81 0724 63 2650 +81 0724 64 9439 Fax: Email: 820838@so.fujioil.jp Mr Susumu Hirano National Association of Food Industry Morinaga & Co. Ltd. 33-1, Shiba 5-Chome, Minato-ku Tokyo 108-8403 **Phone:** +81 3 3456 0122 Fax: +81 3 3456 0536 Email: s-hirano-fd@morinaga.co.jp Mr Takashi Fujita National Association of Food Industry 4-6-5 Tagawa Yodogawa-ku Osaka **Phone:** +81 6 6477 8367 +81 6 6477 6604 Fax: Email: t-fujita@netl.kdd.fr KOREA, REPUBLIC OF CORÉE, RÉPUBLIQUE DE COREA, REPÚBLICA DE Dr. Jin-Hwan Hong Senior Researcher Korea Food and Drug Administration 5 Nokbun-dong, Eunpyung-gu Seoul 122-704 **Phone:** +82 2 380 1666 Fax: +82 2 382 4892 Email: hongjinh@kfda.go.kr Dr. Mee-Ah Park Researcher Ministry of Health and Welfare 5, Nokbun-Dong, Eunpyung-gu Seoul 122-704

 Phone:
 +82 2 380 1559

 Fax:
 +82 2 383 8321

 Email:
 codexkorea@kfda.go.kr

Dr. Man-Chong Lee Manager Lotte group R&D Center Chief Secretary Korea Cacao Chocolate Association 4 Ka-23, Yang Pyung Dong Young Deung Po Gu Seoul **Phone:** +82 2 670 6549 **Fax:** +82 2 6672 6184 **Email:** mclee@lotte.re.kr Mr Chang-Ik Lee Vice President Korea Cacao Chocolate Association 172 Kongneung-2 Dong No Won-GU Seoul Phone: +82 2 784 2309 Fax: +82 2 784 1154 Mr Gun-Hee Han Vice President Korea Cacao & Chocolate Association 316, Faculty of Applied chemical Engineering 172 Kongneung-2 Dong No Won-GU Seoul **Phone:** +82 2 970 6691 +82 2 973 9149 Fax: Email: copac@unitel.co.kr MALAYSIA MALAISIE MALASIA Mr Ahmad Bin Loman Principal Assistant Secretary Ministry of Primary Industries 6-8th Floor, Menara Dayabumi Jalan Sultan Hishamuddin 50654 Kuala Lumpur **Phone:** + 603 2274 7511 Fax: + 603 2274 5649 **Email:** ahmadl@kpu.gov.my Mr Mohammad Jaaffar Ahmad **Regional Manager** Malaysian Palm Oil Board Brickendonbury Hertford, Hertfordshire SG13 8NL England **Phone:** + 44 1992 554 347 Fax: + 44 1992 500564 Email: mpob@mpob.powernet.co.uk Mr Jumali Suratman **Research Officer** Malaysian Cocoa Board Lot 3, Jalan P/9B Seksyen 13 43650 Bandar Baru Bangi Selangor **Phone:** + 603 89267800 Fax: + 603 89255386

Email: jumali@koko.gov.my

MOROCCO MAROC MARRUECOS

M. Mostapha Katifa Ingénieur d'état en IAA Laboratoire Officiel d'Analyses et de Recherches Chimiques Ministère de l'Agriculture, du Developpement rural et des Eaux et Forets 25, Rue Nichakra Rahal Casablanca Phone: + 212 230 21 96/98 Fax: + 212 230 1972 Email: 1.o.a.r.c@casanet.ma

NETHERLANDS PAYS-BAS PAÍSES BAJOS

Mr Eelco Klein Ministry of Agriculture, Nature Management and Fisheries PO Box 20401 2500 EK The Hague Phone: +31 70 378 4252 Fax: +31 70 378 6123 Email: e.klein@ih.agro.nl

Mr Han Zijderveld Manager Laboratories Gerkens Cacao B.V. PO Box 82 1530 AB Wormer Phone: +31 75 646 6161 Fax: +31 75 621 96 26 Email: han_zijderveld@cargill.com

Dr. Marieke Lugt Dutch Biscuit, Chocolate and Confectionery Association Bankastraat 131 B 2585 EL Den Haag Phone: +31 70 355 4700 Fax: +31 70 358 4679 Email: vnfkd@vbz.nl

Mr Aloys Velthuis Legal Manager Mars/Masterfoods Europe Taylorweg 5 5466AE Veghel Phone: +31 413 383482 Fax: +31 413 383855 Email: aloys.velthuis@eu.effem.com M. Fons Kasbergen Vice-Président de CAOBISCO Rue Defacqz, 1 B-1000 Brussels, Belgium **Phone:** +32 2 539 18 00 Fax: +32 2 539 15 75 Email: fons.kasbergen@eu.effem.com SPAIN **ESPAGNE ESPAÑA** Dr. Elisa Revilla García Jefe de Área de Coordinación Sectorial de la Subdirección General de Planificación Alimentaria Dirección General de Alimentación del Ministerio de Agricultura, Pesca y Alimentación Paseo Infanta Isabel 1 E-28071 Madird **Phone:** +34 91 347 45 96 Fax: +34 91 347 57 28 Email: erevilla@mapya.es D. Jorge Carbonnell Food Chemist Grupo NUTREXPA S.A. c/Lepant 410-414 E-08025 Barcelona **Phone:** +34 93 290 02 75 Fax: +34 93 290 02 83 Email: carbonell@nutrexpa.es **SWEDEN** SUÈDE **SUECIA** Mrs Kerstin Jansson Senior Administrative Officer Ministry of Agriculture S-103 33 Stockholm **Phone:** +46 8 405 1168 +46 8 206 496 Fax: **Email:** kerstin.jansson@agriculture.ministry.se Mrs Eva Lönberg Codex Coordinator for Sweden National Food Administration Box 622 S-751 26 Uppsala **Phone:** +46 18 17 55 47 +46 18 10 58 48 Fax: Email: eva.lonberg@slv.se

SWITZERLAND SUISSE SUIZA

Mr Pierre Studer Food and Consumer Safety Swiss Federal Office of Public Health CH-3003 Berne Phone: +41 31 3233105 Fax: +41 31 3229574 Email: pierre.studer@bag.admin.ch Mrs Awilo Ochieng-Pernet Swiss Federal Office of Public Health 3003 Berne, Switzerland Phone: +41 31 3220041 Fax: +41 31 3229574 Email: awilo.ochieng@bag.admin.ch

Mr Kurt Hunzinger Secretary Chocosuisse Postfach 1007 3000 Bern 7 Phone: +41 31 310 09 90 Fax: +41 31 310 09 99 Email: kurt.hunzinger@chocosuisse.ch

Mr Laurent Farinelli Nestec SA 1800 Vevey Phone: +41 21 924 42 10 Fax: +41 21 924 45 47 Email: laurent.farinelli@nestle.com

Mr Markus Lutz Chocolat Bernrain AG Bündtstrasse 12 8280 Kreuzlingen Phone: +41 71 677 97 77 Fax: +41 71 677 97 50 Email: mlutz@swisschocolate.ch

THAILAND THAÏLANDE

TAILANDIA

Mrs Phani Na Rangsi Thai Industrial Standards Institute Ministry of Industry Rama VI Rd. Bangkok 10400 **Phone:** +662 573 3048 **Fax:** +662 248 7987 Mrs Sasiwimon Tabyam Standards Officer 5 Thai Industrial Standards Institute Ministry of Industry Rama VI Rd. Bangkok 10400 Phone: +662 246 1993 Fax: +662 248 7987 Email: siwimont@tisi.go.th Ms Kakanang Krajangwong Ministry of Foreign Affairs Royal Thai Embassy in Bern Kirchstrasse 56 CH-3097 Liebefeld Phone: +41 31 970 30 30 Fax: +41 31 970 30 35

TOGO

Mr. A. Outouloum Sambo Directeur de cabinet Ministère de l'Industrie du Commerce et des Transports B.P. 383 Lomé Phone: +228 21 29 84 Fax: +228 21 05 72

UNITED KINGDOM ROYAUME-UNI REINO UNIDO

Mr. Grant Meekings Head Food Labelling Standards & Consumer Protective Division Food Standards Agency P.O. Box 31037 London SW1P 3WG Phone: +44 20 7238 6278 Fax: +44 20 7238 6763 Email:grant.meekings@foodstandards.gsi.go v.uk

Mr. Mark Woolfe Head of Branch Food Labelling, Standards and Consumer Protection Division Food Standards Agency P.O. Box 31037 London SW1P 3WG Phone: +44 20 7238 6168 Fax: +44 20 7238 6763 Email: mark.woolfe@foodstandards.gsi.gov.uk Ms. Wendy Hart Statutory and Standards Manager Cadbury Ltd. Bournville Birmingham B30 2LU Phone: +44 121 458 2000 Fax: +44 121 451 4333 Email: Wemdy-Hatth@csplc.com

Mrs.Morag Bullard Legal Affairs Scientist Mars UK Dundee RD Slough, Berks Phone: +44 1753 514261 Fax: +44 1753 514737 Email: morag.bullard@eu.effem.com

Mr. Michael Webber Scientific & Regulatory Manager Biscuit Cake Chocolate & Confectionery Alliance 37-41 Bedford Row London WC1R 4JH Phone: +44 207 404 9115 Fax: +44 207 404 9139 Email: mike@becca.org.uk

UNITED S TATES OF AMERICA ETATS - UNIS D'AMÉRIQUE ESTADOS UNIDOS DE AMÉRICA

Mr Charles W. Cooper Director, International Activities Staff Food and Drug Administration Center for Food Safety and Applied Nutrition 200 C Street, SW (HFS-585) Washington, DC 20204 Phone: +1 202 205 5042 Fax: +1 202 401 7739 Email: charles.cooper@cfsan.fda.gov

Dr. Michelle A. Smith Food Scientist (HFS-306) Food and Drug Administration Center for Food Safety and Applied Nutrition 200 C Street, S.W. Washington, D.C. 20204 Phone: +1 202 205 2975 Fax: +1 202 205 4422 Email: msmith1@cfsan.fda.gov

Ms Shellee Anderson Interdisciplinary Scientist Office of Plant and Dairy Foods and Beverages Food and Drug Administration U.S. Food and Drug Administration 200 C Street, SW Washington, DC 20204 Phone: +1 202 205 5023 Fax: +1 202 205 4422 Email: shellee.anderson@cfsan.fda.gov Dr. Dennis Keefe Special Assistant Office of Premarket Approval (HFS-200) Food and Drug Administration Center for Food Safety and Applied Nutrition 200 C Street, S.W. Washington, D.C. 20204 **Phone:** +1 202 418 3113 +1 202 418 3131 Fax: **Email:** dkeefe@cfsan.fda.org Ms Ellen Matten U.S. Codex Office Food Safety and Inspection Service U.S. Department of Agriculture 1400 Independence Avenue, SW Room 4861 South Building Washington DC 20520-3700 **Phone:** +1 202 205 7760 +1 202 720 3157 Fax: Email: ellen.matten@usda.gov Dr. Raymond C. Glowaky Senior Vice President, Scientific Affairs American Cocoa Research Institute 7900 Westpark Ave, Suite A-320 McLean, VA 22102 **Phone:** +1 703 790 5011 +1 703 790 0168 Fax: Email: rglowaky@candyusa.org Dr. Steve Rizk Manager, Scientific + Regulatory Affairs M&M Mars 800 High Street Hackettstown, NJ 07840 **Phone:** +1 908 850 2753 +1 908 850 2697 Fax: Email: steve.rizk@effem.com Dr. Stanley M. Tarka, Jr. Senior Director Food Science & Technology Hershey Foods Corporation **Technical Center** P.O. Box 805 Hershev, PA 17033 **Phone:** +1 717 534 5130 +1 717 534 5076 Fax

Email: starka@hersheys.com

REPUBLICA BOLIVARIANA DE VENEZUELA

Mrs. Ofelia Herrera, M.S. Biologist, Head Research Section of Direction of Food Hygiene Ministry of Health and Social Development Centro Semón, Edif. Sur, ala Oeste Piso 3, ofc. 313 Caracas 1010 Phone: +58-2 484 3066 or 58-2 372 5068 Fax: +58-2 484 3066 014 204 2228

Email: abebizz@aol.com

INTERNATIONAL GOVERNMENTAL ORGANIZATIONS

ALLIANCE DES PAYS PRODUCTEURS DE CACAO COCOA PRODUCERS' ALLIANCE (COPAL)

Mr Hope Sona Ebai Secretary General Cocoa Producers Alliance COPAL Western House (11th Floor) 8/10 Broad Street P.O.Box 1718 Lagos, Nigeria **Phone:** +2341 263 5574 **Fax:** +2341 263 5684 **Email:** copal@alpha.linkserve.com

COUNCIL OF THE EUROPEAN UNION

Mr Olli Mattila Administrator General Secretariat of the Council of the European Union 175 Rue de la Loi 1048 Brussels Belgium Phone: +32 2 285 8357 Fax: +32 2 285 7928 Email: olli.mattila@consilium.eu.int

EUROPEAN COMMISSION

Mr Alberik Scharpe Deputy Head of Unit European Commission AGRI-B-4 Rue de la Loi 200 1040 Brussels Belgium Phone: +32 2 295 59 57 Fax: +32 2 296 59 63 Email: alberik.scharpe@cec.eu.int Mr Jean-François Roche Administrator European Commission Health and Consumer Protection Directorate General Frossart 101 200, rue de la Loi B- 1069 Bruxelles Phone: +32 2 295 31 41 Fax: +32 2 296 09 51 Email: jean-francois.roche@cec.eu.int

INTERNATIONAL NON-GOVERNMENTAL ORGANIZATIONS

COMITÉ EUROPÉEN DES FABRIQUANTS DE SUCRE (CEFS)

Dr Nathalie Henin Conseiller scientifique comité Européen des Fabricants de Sucre 182, avenue de Tervuren 1150 Bruxelles, Belgium Phone: +32 2 774 51 06 Fax: +32 2 771 00 26 Email: nathalie.henin@cefs.org

EUROPEAN FOOD LAW ASSOCIATION (EFLA)

Mr Guy Valkenborg Board member and Treasurer European Food Law Association EFLA 235, rue de la Loi, Brussels B-1000 Brussels, Belgium Phone: +32 2 218 14 70 Fax: +32 2 219 73 42 Email: guyvalkenborg@eas.be

Mr Mikael, Conny Svensson European Food Law Association EFLA 235, rue de la Loi, Brussels B-1000 Brussels, Belgium Phone: +32 2 218 14 70 Fax: +32 2 219 73 42

INTERNATIONAL OFFICE OF COCOA, CHOCOLATE AND SUGAR CONFECTIONERY (IOCCC)

Mr Tom Harrison Président de l'OICCC Herons Court Station Road Wargrave Berks United Kingdom Phone: +44 011 8940 2138 Fax: +44 011 8940 2138 Mme Pénélope Alexandre Manager Technical and Legislative affairs IOCCC Rue Defacqz, 1 B-1000 Brussels, Belgium Phone: +32 2 539 18 00 Fax: +32 2 539 15 75 Email: penelope.alexandre@caobisco.be

M. Fons Kasbergen Vice-Président de CAOBISCO Rue Defacqz, 1 B-1000 Brussels, Belgium Phone: +32 2 539 18 00 Fax: +32 2 539 15 75 Email: fons.kasbergen@eu.effem.com

M. David Zimmer Secrétaire Général de CAOBISCO Rue Defacqz, 1 B-1000 Brussels, Belgium Phone: +32 2 539 18 00 Fax: +32 2 539 15 75 Email: david.zimmer@caobisco.be

EUROPEAN COCOA ASSOCIATION

Mr Robert A. Zehnder General Secretary European Cocoa Association Av. De Cortenbergh 118 P.O. Box 8 B-1000 Brussels, Belgium Phone: +32 2 237 95 75 Fax: +32 2 237 95 74 Email: robert.a.zehnder@cocoa.eu.com

JOINT FAO/WHO SECRETARIAT

Dr Alan W. Randell Secretary, Codex Alimentarius Commission Joint FAO/WHO Food Standards Programme Food and Nutrition Division Food and Agriculture Organization of the United Nations Viale delle Terme di Caracalla 00100 Rome, Italy Phone: +39 06 570.54390 Fax: +39 06 570.54593 Email: alan.randell@fao.org Mr Christophe Leprêtre Associate Food Standards Officer Joint FAO/WHO Food Standards Programme Food and Nutrition Division Food and Agriculture Organization of the United Nations Viale delle Terme di Caracalla 00100 Rome, Italy Phone: +39 06 570.55621 Fax: +39 06 570.54593 Email: christophe.lepretre@fao.org

APPENDIX II

DRAFT REVISED STANDARD FOR COCOA BUTTER

(Advanced at Step 8 of the Procedure)

1 SCOPE

This standard applies exclusively to cocoa butter used as ingredient in the manufacture of chocolate and chocolate products.

2 DESCRIPTION

2.1 DEFINITION OF COCOA BUTTER

Cocoa butter is the fat obtained from cocoa beans with the following characteristics:

- free fatty acid content (expressed as oleic acid):	not more than 1.75% m/m
- unsaponifiable matter:	not more than 0.7% m/m, except in the case of press cocoa butter which shall not be more than 0.35% m/m
3 FOOD ADDITIVES	
3.1 PROCESSING AID	
	Maximum Level
Hexane (62°C - 82°C)	1 mg/kg
4 CONTAMINANTS	
4.1 HEAVY METALS	
	Maximum Level
Lead (Pb)	0.1 mg/kg

5 HYGIENE

5.1 It is recommended that the products covered by the provisions of this standard be prepared in accordance with the appropriate sections of the *Recommended International Code of Practice - General Principles of Food Hygiene* — CAC/RCP 1-1969, Rev. 3-1997.

5.2 To the extent possible in good manufacturing practice, the products shall be free from objectionable matter.

5.3 When tested by appropriate methods of sampling and examination, the product shall not contain any substances originating from microorganisms in amounts which may represent a hazard to health.

5.4 When tested by appropriate methods of sampling and analysis, the products shall be free from microorganisms in amounts which may represent a hazard to health.

6 LABELLING

In addition to the *Codex General Standard for Labelling of Prepackaged Foods* — CODEX STAN 1-1985, Rev.1-1991 — the following apply:

6.1 NAME OF THE PRODUCT

6.1.1 Cocoa Butter

The product shall be '*Cocoa Butter*'' except that the name '*Press Cocoa Butter*'' may be used for products in conformity with the description for this product given in Section 2.1.

6.2 LABELLING OF NON-RETAIL CONTAINERS

Information required in Section 6.1 of this Standard and Section 4 of the Codex General Standard for the Labelling of Prepackaged Foods shall be given either on the container or in accompanying documents, except that the name of the product, lot identification, and the name and address of the manufacturer or packer shall appear on the container. However, lot identification, and the name and address of the manufacturer or packer may be replaced by an identification mark provided that such a mark is clearly identifiable with the accompanying documents.

7 METHODS OF ANALYSIS AND SAMPLING

7.1 DETERMINATION OF FREE FATTY ACIDS

According to IUPAC (1987) 2.201.

7.2 DETERMINATION OF UNSAPONIFIABLE MATTER

According to IUPAC (1987) 2.401.

7.3 DETERMINATION OF LEAD

According to AOAC 934.07 or IUPAC Method (Pure & Appl. Chem., 63).

APPENDIX III

DRAFT REVISED STANDARD FOR COCOA (CACAO) MASS (COCOA/CHOCOLATE LIQUOR) AND COCOA CAKE

(Advanced at Step 8 of the Procedure)

1 SCOPE

This standard applies to Cocoa (Cacao) Mass or Cocoa/Chocolate Liquor, and Cocoa Cake, as defined, for the use in the manufacture of cocoa and chocolate products. These products may also be sold directly to the consumer.

2 DESCRIPTIONS

2.1 COCOA (CACAO) MASS (COCOA/CHOCOLATE LIQUOR)

Cocoa (Cacao) Mass or Cocoa/Chocolate Liquor is the product obtained from cocoa nib, which is obtained from cocoa beans of merchantable quality which have been cleaned and freed from shells as thoroughly as is technically possible with/without roasting, and with/without removal or addition of any of its constituents.

2.2 COCOA CAKE

Cocoa Cake is the product obtained by partial or complete removal of fat from cocoa nib or cocoa mass.

3 ESSENTIAL COMPOSITION AND QUALITY FACTORS

3.1 COCOA MASS (COCOA/CHOCOLATE LIQUOR)

Cocoa Shell and Germ	not more than 5% m/m calculated on the fat-free dry matter		
	or		
	not more than 1.75% calculated on an alkali free basis (for Cocoa Shell only)		
Cocoa Butter	47-60% m/m		

3.2 COCOA CAKE

Cocoa Shell and Germ	not more than 5% m/m calculated on the fat-free dry matter.		
	or		
	not more than 1.75% calculated on an alkali free basis (for Cocoa Shell only)		
Cocoa Butter	47-60% m/m		

.

4 FOOD ADDITIVES

Only those food additives listed below may be used and only within the limits specified.

Maximum Level

4.1 ACIDITY REGULATORS

503(i)	Ammonium carbonate		
527	Ammonium hydroxide		
503(ii)	Ammonium hydrogen carbonate		
170(i)	Calcium carbonate		
330	Citric acid		
504(i)	Magnesium carbonate		
528	Magnesium hydroxide		
530	Magnesium oxide		
501(i)	Potassium carbonate	Limited by GMP	
525	Potassium hydroxide		
501(ii)	Potassium hydrogen carbonate		
500(i)	Sodium carbonate		
524	Sodium hydroxide		
500(ii)	Sodium hydrogen carbonate		
526	Calcium hydroxide		
338	Orthophosphoric acid	2.5 g/kg expressed as P ₂ 0 ₅ in finished cocoa and chocolate products	
334	L-Tartaric acid	5 g/kg in finished cocoa and chocolate products	
4.2 EM	IULSIFIERS	Maximum Level	
471	Mono- and diglycerides of edible fatty acids	Limited by GMP	
322	Lecithin	Linined by GMP	
442	Ammonium salts of phosphatidic acids	10 g/kg in finished cocoa or chocolate products	
476	Polyglycerol esters of interesterified ricinoleic acid	5 g/kg in finished cocoa or chocolate products	

4.3	FLAVOURING AGENTS	Maximum Level
	Natural and artificial flavours, except those which reproduce the flavour of chocolate or milk Vanillin Ethyl vanillin	Limited by GMP
5	CONTAMINANTS	
5.1	HEAVY METAL	Maximum Level

Lead

1 mg/kg

6 HYGIENE

- **6.1** It is recommended that the products covered by the provisions of this Standard be prepared in accordance with the appropriate sections of the *Recommended International Code of Practice General Principles of Food Hygiene* CAC/RCP 1-1969, Rev. 3-1997.
- **6.2** To the extent possible in good manufacturing practice, the products shall be free from objectionable matter.
- **6.3** When tested by the appropriate methods of sampling and examination, the products:

(a) shall be free from microorganisms capable of development under normal conditions of storage; and

(b) shall not contain any substances originating from microorganisms in amounts which may represent a hazard to health.

7 LABELLING

In addition to the *Codex General Standard for Labelling of Prepackaged Foods* - CODEX STAN 1-1985, Rev.1-1991 - the following apply:

7.1 NAME OF THE PRODUCT

7.1.1 The name used to describe the product defined under Section 2.1 and complying with provisions of Section 3.1 of the Standard include "Cocoa (Cacao) Mass", "Cocoa/Chocolate Liquor", "Cocoa Paste", "Unsweetened Chocolate" and "Bitter Chocolate".

In the case of products traded internationally, the name of the product shall be acceptable to the importing authorities.

7.1.2 The name of the product described under Section 2.2 and complying with provisions of Section 3.2 of the Standard shall be: *"Cocoa Cake"*.

7.2 LABELLING OF NON-RETAIL CONTAINERS

Information required in Section 7.1 of this Standard and Section 4 of the Codex General Standard for the Labelling of Prepackaged Foods shall be given either on the container or in accompanying documents, except that the name of the product, lot identification, and the name and address of the manufacturer or packer shall appear on the container. However, lot identification, and the name and address of the manufacturer or packer may be replaced by an identification mark provided that such a mark is clearly identifiable with the accompanying documents.

8 METHODS OF ANALYSIS AND SAMPLING

8.1 DETERMINATION OF COCOA SHELL

According to AOAC 968.10 and 970.23.

8.2 DETERMINATION OF FAT CONTENT

According to AOAC 963.15 or IOCCC 14 (1972).

8.3 DETERMINATION OF LEAD

According to AOAC 986.15 (Codex general method).

APPENDIX IV

DRAFT REVISED STANDARD FOR COCOA POWDERS (COCOAS) AND DRY MIXTURES OF COCOA AND SUGARS

(Advanced at Step 8 of the Procedure)

1 SCOPE

This standard applies to cocoa powders (cocoas) and dry mixtures of cocoa and sugars intended for direct consumption.

2 **DESCRIPTION**

2.1 COCOA PRODUCTS

2.1.1 "*Cocoa Powder*" and "*Fat-reduced Cocoa Powder*" and "*Highly Fat-reduced Cocoa Powder*" are the products obtained from cocoa cake (*Cf.* Standard for Cocoa (Cacao) Mass (Cocoa/Chocolate Liquor) and Cocoa Cake) transformed into powder.

3 ESSENTIAL COMPOSITION AND QUALITY FACTORS

3.1 ESSENTIAL COMPOSITION

3.1.1 Moisture Content

Not more than 7% m/m

3.1.2 Cocoa Powder and Dry Mixtures of Cocoa and Sugars

		COCOA BUTTER CONTENT (as a minimum cocoa powder content on a dry matter basis)			
		^з 20% m/m	³ 10% m/m but < 20% m/m	< 10% m/m	
Сосоа	POWDERS ONLY	Cocoa Powder	Fat-Reduced Cocoa Powder	Highly Fat reduced Cocoa Powder	
×.	Not < 25% m/m	Sweetened Cocoa,	Sweetened Cocoa, Fat- reduced,	Sweetened Cocoa, Highly Fat-reduced	
COCOA POWDER CONTENT IN DRY MIXTURES		<u>or</u> Sweetened Cocoa Powder,	or Sweetened Cocoa Powder, Fat-reduced,	<u>or</u> Sweetened Cocoa Powder, Highly Fat-reduced	
ENT IN DRY		<u>or</u> Drinking Chocolate	<u>or</u> Fat-Reduced Drinking Chocolate	or Highly Fat- Reduced Drinking Chocolate	
ER CONT	Not < 20% m/m	Sweetened Cocoa Mix,	Sweetened Cocoa Mix, Fat-reduced,	Sweetened Cocoa Mix, Highly Fat- reduced	
COA POWD		or Sweetened Mixture with Cocoa	or Sweetened Mixture with Cocoa, Fat-reduced:	<u>or</u> Sweetened Mixture with Cocoa, Highly Fat-reduced	
CO	< 20% m/m	Sweetened Cocoa- flavoured Mix	Sweetened Cocoa- flavoured Mix, Fat- reduced	Sweetened Cocoa- flavoured Mix, Highly Fat-reduced	

3.1.3 Chocolate Powder

Chocolate Powder: mixture of cocoa powder and sugars and/or sweeteners, containing not less than 32% m/m cocoa powder (29% m/m on a dry matter basis).

3.2 OPTIONAL INGREDIENTS

Spices

Salt (Sodium chloride)

4 FOOD ADDITIVES

4 FUC	JD ADDITIVES	
	CIDITY REGULATORS	Maximum level
503(i)	Ammonium carbonate	
503(ii)	Ammonium hydrogen carbonate	
527	Ammonium hydroxide	
170(i)	Calcium carbonate	
330	Citric acid	
504(i)	Magnesium carbonate	
528	Magnesium hydroxide	Limited by GMP
530	Magnesium Oxide	
501(i)	Potassium carbonate	
501(ii)	Potassium hydrogen carbonate	
525	Potassium hydroxide	
500(i)	Sodium carbonate	
500(ii)	Sodium hydrogen carbonate	
524 526	Sodium hydroxide	
526	Calcium hydroxide	
338	Orthophosphoric acid	2.5 g/kg, expressed as P_2O_5 , on the cocoa fraction of the finished product
334	L-Tartaric acid	5 g/kg on the cocoa fraction of the finished product
4.2 EN	IULSIFIERS	
471	Mono- and di-glycerides of edible	
	fatty acids	Limited by GMP
322	Lecithin	
476	Polyglycerol esters of interesterified	$5 \mathrm{g/kg}$ on the finished product
., 0	ricinoleic acid	
442	Ammonium salts of phosphatidic acids	10 g/kg on the finished product
473	Edible sucrose esters of fatty acids ¹	10 g/kg on the finished product
4.3 FL	AVOURING AGENTS	Maximum Level
	tural and artificial flavours, except	
	se which reproduce the flavour of	Limited by GMP
cho	colate or milk	
Vai	nillin	Limited by GMP
Eth	yl vanillin	
4.4 AN	TI-CAKING AGENTS	Maximum Level
559	Aluminium silicate	
552	Calcium silicate	
553(i)	Magnesium silicate	
553(ii)	Magnesium trisilicate	10 g/kg on Dry Mixtures of Cocoa and Sugars
553(iii)	Talc	
554	Sodium aluminosilicate	
551	Silicon dioxide, amorphous	
341(iii)	Tricalcium phosphate	l

¹ Temporarily endorsed.

4.5 BULKING AGENT	Maximum Level
1200 Polydextroses A and N	Limited by GMP
4.6 SWEETENERS	Maximum Level
953 Isomalt (Isomaltitol)	
966 Lactitol	
421 Mannitol	Limited by GMP
965 Maltitol and maltitol syrup	
420 Sorbitol and sorbitol syrup	
967 Xylitol	
950Acesulfame Potassium	350 mg/kg on finished product
951 Aspartame	3 g/kg on finished product
955 Sucralose	580 mg/kg on finished product
954 Sodium Saccharin	100 mg/kg: residue limit on finished product
957 Thaumatin	Limited by GMP on finished product
5 CONTAMINANTS	
5.1 HEAVY METAL	
	Maximum level on the Cocoa Fraction
Lead	1 mg/kg

6 HYGIENE

6.1 It is recommended that the products covered by the provisions of this standard be prepared in accordance with the appropriate sections of the *Recommended International Code of Practice - General Principles of Food Hygiene* (CAC/RCP 1-1969, Rev. 3-1997).

6.2 To the extent possible in good manufacturing practice, the products shall be free from objectionable matter.

6.3 When tested by appropriate methods of sampling and examination, the products shall not contain any substances originating from microorganisms in amounts which may represent a hazard to health.

6.4 When tested by appropriate methods of sampling and analysis, the products shall be free from microorganisms in amounts which may represent a hazard to health.

7 LABELLING

In addition to the provisions of the *Codex General Standard for Labelling of Prepackaged Foods* — CODEX STAN 1-1985, Rev.1-1991 — the following apply:

7.1 NAME OF THE FOOD

7.1.1 Cocoa Powders and Dry Mixtures of Cocoa and Sugars

The descriptions of the products defined in Section 3.1.2 shall be used as the name of the food in each case.

7.1.2 Chocolate powder

The name of the product complying with Section 3.1.3 of the standard shall be: *'Chocolate Powder'* (*"Chocolat en poudre"*).

7.1.3 The words 'minimum cocoa powder (or fat-reduced cocoa powder or highly fat-reduced cocoa powder) content x%' shall be declared in accordance with dispositions of Section 5.1.1 of the Codex General Standard for Labelling of Prepackaged Foods, where required by national legislation.

7.1.4 Sweeteners

When sweeteners described in section 4.6 have been used to replace sugars, totally or partially, the word *"sugared"*, or equivalent, shall not appear in the name of the food: according to the language of the importing country, the term *"sweetened"* or equivalent shall be used.

Where sweeteners have been added to the product, the term "*artificially sweetened*" or "*with added sweeteners*" or "*with sugars and added sweeteners*" shall appear in close proximity to the name of the food.

7.1.5 National laws should only permit the use of names other than those given in 7.1.1 and 7.1.2. in countries where such names are traditional, fully understood by the consumer and not misleading or deceptive with respect to other categories of product provided that any product with less than 25% of total cocoa powder or fat-reduced cocoa powder content shall not bear the term "chocolate" in its designation.

8 METHODS OF ANALYSIS AND SAMPLING

8.1 DETERMINATION OF COCOA BUTTER

According to IOCCC 37 (1990) and IOCCC 14 (1972).

8.2 DETERMINATION OF COCOA POWDER, FAT REDUCED COCOA POWDER AND HIGHLY FAT-REDUCED COCOA POWDER

To be developed

8.3 DETERMINATION OF MOISTURE

According to AOAC 977.04 or IOCCC 26 (1988)-Karl Fisher Method.

8.4 DETERMINATION OF LEAD

According to AOAC 934.07.

APPENDIX V

PROPOSED DRAFT STANDARD FOR CHOCOLATE AND CHOCOLATE PRODUCTS

(Advanced at Step 5 of the Procedure)

1 SCOPE

The standard applies to all chocolate and chocolate products for sale to consumers. Chocolate and chocolate products shall be prepared from cocoa and cocoa materials with sugars and may contain sweeteners, milk products, flavouring substances and other food ingredients.

2 DESCRIPTION AND ESSENTIAL COMPOSITION FACTORS

2.1 CHOCOLATE TYPES

Chocolate is the generic name for the homogenous products complying with the descriptions below and summarized in Table 1. It is obtained by an adequate manufacturing process from cocoa materials which may be combined with milk products, sugars and/or sweeteners, emulsifiers and/or flavours. Other edible foodstuffs, excluding added flour and starch (except for products in 2.4.3 of this Standard) and animal fats other than milk fat, may be added to form various chocolate products. These combined additions shall be limited to 40 % of the total weight of the finished product, subject to the labelling provisions under Section 6.

The addition of vegetable fats other than cocoa butter shall not exceed 5% of the finished product, after deduction of the total weight of any other added edible foodstuffs, without reducing the minimum contents of cocoa materials. Where required by the authorities having jurisdiction, the nature of the vegetable fats permitted for this purpose may be prescribed in applicable legislation.

2.1.1 Chocolate

Chocolate (or *bittersweet chocolate, semi-sweet chocolate*, or *dark chocolate*), shall contain, on a dry matter basis, not less than 35% total cocoa solids, of which not less than 18% shall be cocoa butter and not less than 14% fat-free cocoa solids.

2.1.2 Sweet Chocolate

Sweet Chocolate shall contain, on a dry matter basis, not less than 30% total cocoa solids, of which at least 18% shall be cocoa butter and at least 12% fat-free cocoa solids.

2.1.3 Couverture Chocolate

Couverture Chocolate shall contain, on a dry matter basis, not less than 35% total cocoa solids of which not less than 31% shall be cocoa butter and not less than 2.5% of fat-free cocoa solids.

2.1.4 Milk Chocolate types

2.1.4.1 Milk Chocolate

Milk Chocolate shall contain, on a dry matter basis, not less than 25% cocoa solids (including a minimum of 2.5% fat-free cocoa solids) and a specified minimum of milk solids, referring to the addition of milk ingredients in their natural proportions except that milk fat may be added or removed, between 12% and 14% (including a minimum of milk fat between 2.5% and 3.5%) as applied by the authority having jurisdiction in accordance with applicable legislation.

2.1.4.2 Family Milk Chocolate

Family Milk Chocolate shall contain, on a dry matter basis, not less than 20% cocoa solids (including a minimum of 2.5% fat-free cocoa solids) and not less than 20% milk solids, referring to the addition of milk ingredients in their natural proportions except that milk fat may be added or removed, (including a minimum of 5% milk fat).

2.1.4.3 Milk Couverture Chocolate

Milk Couverture Chocolate shall contain, on a dry matter basis, not less than 25 % cocoa solids (including a minimum of 2.5% non-fat cocoa solids) and not less than 14% milk solids, referring to the addition of milk ingredients in their natural proportions except that milk fat may be added or removed, (including a minimum of 3.5% milk fat) and a total fat of not less than 31%.

2.1.5 Cocoa Butter Confectionery / White Chocolate

Cocoa Butter Confectionery / White Chocolate shall contain, on a dry matter basis, not less than 20% cocoa butter and not less than 14% milk solids, referring to the addition of milk ingredients in their natural proportions except that milk fat may be added or removed, (including a minimum milk fat in a range of 2.5% to 3.5% as applied by the authority having jurisdiction in accordance with applicable legislation).

2.2 VERMICELLI AND FLAKES (DROPS, SHAVINGS)

Vermicelli and Flakes (Drops, Shavings) are cocoa products obtained by a mixing, extrusion and hardening technique which gives unique, crisp textural properties to the products. Vermicelli are presented in the form of short, cylindrical grains and flakes in the form of small flat pieces.

2.2.1 Chocolate Vermicelli / Flakes (Drops, Shavings)

Chocolate Vermicelli / Flakes (Drops, Shavings) shall contain, on a dry matter basis, not less than 32% total cocoa solids, of which at least 12% shall be cocoa butter and 14% fat-free cocoa solids.

2.2.2 Milk Chocolate Vermicelli / Flakes (Drops, Shavings)

Milk Chocolate Vermicelli / Flakes (Drops, Shavings) shall contain, on a dry matter basis, not less than 20% cocoa solids (including a minimum of 2.5% fat-free cocoa solids) and not less than 12% milk solids, referring to the addition of milk ingredients in their natural proportions except that milk fat may be added or removed, (including a minimum of 3% milk fat).

2.3 FILLED CHOCOLATE

Filled Chocolate is a product covered by a coating of one or more of the Chocolates defined in Section 2.1 or 2.4, with exception of chocolate a la taza, of this Standard, , the centre of which is clearly distinct through its composition from the external coating. Filled Chocolate does not include Flour Confectionery, Pastry, Biscuit or Ice Cream products. The chocolate part of the coating must make up at least 25% of the total weight of the product concerned.

If the centre part of the product is made up of a component or components for which a separate Codex Standard exists, then the component(s) must comply with this applicable standard.

2.4 OTHER CHOCOLATE PRODUCTS

Other chocolate products are the additional commercially available products that rely for their essential character either totally or extensively on cocoa materials.

2.4.1 A Chocolate or Praline ("Bonbon de Chocolat" or "Praline")

A *Chocolate* or *Praline* designates the product in a single mouthful size, where the amount of the chocolate component shall not be less than 25% of the total weight of the product. The product shall consist of either filled chocolate or a single or combination of the chocolates as defined under Sections 2.1 and 2.4, with exception of *chocolate a la taza*.

2.4.2 Gianduja Chocolates

2.4.2.1 Gianduja Chocolate ("Chocolat aux noisettes Gianduja")

Gianduja chocolate is the product obtained, firstly, from chocolate having a minimum total dry cocoa solids content of 32%, including a minimum dry non-fat cocoa solids content of 8%, and, secondly, from finely ground hazelnuts, in such quantities that 100 g of the product contain not less than 20 g of hazelnuts.

2.4.2.2 Milk Gianduja Chocolate ("Chocolat au lait et aux noisettes Gianduja")

Milk Gianduja Chocolate is the product obtained, firstly, from milk chocolate having a minimum dry milk solids, referring to the addition of milk ingredients in their natural proportions except that milk fat may be added or removed, content of 10% and, secondly, from finely ground hazelnuts such that 100 g of the product contain not less than 15 g of hazelnuts.

2.4.3 Chocolate a la taza

2.4.3.1 *Chocolate a la taza* is the product described under Section 2.1.1 of this Standard and containing a maximum of 8% m/m flour and/or starch.

2.4.3.2 *Chocolate familiar a la taza* is the product described under Section 2.1.2 of this Standard and a maximum of 18% m/m flour and/or starch.

TABLE 1 (1/2): SUMMARY TABLE OF COMPOSITIONAL REQUIREMENTS OF SECTION 2

(% calculated on the dry matter in the product and after deduction of the weight of the other edible foodstuffs authorized under Section 2.1)

Constituent	rs Cocoa	Fat-free	Total Cocoa	Milk	Total Milk
	Butter	Cocoa	Solids	Fat	Solids
PRODUCTS		Solids			
2.1 CHOCOLATE TYPES					
2.1.1 Chocolate	≥ 18	≥ 14	≥ 35		
2.1.2 Sweet Chocolate	≥ 18	≥ 12	≥ 30		
2.1.3 Couverture Chocolate	≥ 31	≥ 2.5	≥ 35		
2.1.4. Milk Chocolate Types					
2.1.4.1 Milk Chocolate		≥ 2.5	≥ 25	2.5-3.5	12-14
2.1.4.2 Family Milk Chocolate		≥ 2.5	≥ 20	≥ 5	≥ 20
2.1.4.3 Milk Couverture Chocolate		≥ 2.5	≥ 25	≥ 2.5	≥ 14
2.1.5 Cocoa Butter Confectionery / White Chocolate	≥ 20			2.5-3.5	≥ 14
2.2 VERMICELLI AND FLAKES (DROPS, SHAVINGS)					
2.2.1 Chocolate Vermicelli/Flakes (Drops, Shavings)	≥ 12	≥ 14	≥ 32		
2.2.2 Milk Chocolate Vermicelli/Flakes (Drops, Shavings)		≥ 2.5	≥ 20	≥3	≥ 12
2.3 FILLED CHOCOLATE (See Section 2.3)					

TABLE 1 (2/2): SUMMARY TABLE OF COMPOSITIONAL REQUIREMENTS OF SECTION 2

(% calculated on the dry matter in the product and after deduction of the weight of the other edible foodstuffs authorized under Section 2.1)

PRODU	CONSTITUENTS	Cocoa Butter	Fat-free Cocoa Solids	Total Cocoa Solids	Milk Fat	Total Milk Solids
2.4	OTHER CHOCOLATE PRODUCTS					
2.4.1	A Chocolate or Praline (See Section 2.4.1)					
2.4.2	Gianduja Chocolates					
2.4.2.1	Gianduja Chocolate (See Section 2.4.2.1)		≥ 8	≥ 32		
2.4.2.1	Milk Gianduja Chocolate (See Section 2.4.2.2)					≥ 10
2.4.3	Chocolate a la Taza					
2.4.3.1	Chocolate a la taza (See Section 2.4.3.1)					
2.4.3.2	Chocolate Familiar a la taza (See Section 2.4.3.2)					

3 FOOD ADDITIVES

Only those food additives listed below may be used and only within the limits specified.

3.1 Alkalizing and neutralizing agents carried over as a result of processing cocoa materials in proportion to the maximum quantity as provided for.

3.2 ACIDITY REGULATORS			Maxin	um Level	
503(i)	Ammonium carbonate				
527	Ammonium hydroxide				
503(ii)	Ammonium hydrogen carbonate				
170(i)	Calcium carbonate				
330	Citric acid				
504(i)	Magnesium carbonate				
528	Magnesium hydroxide				
530	Magnesium oxide		Limite	d by GMP	
501(i)	Potassium carbonate				
525	Potassium hydroxide				
501(ii)	Potassium hydrogen carbonate				
500(i)	Sodium carbonate				
524	Sodium hydroxide				
500(ii)	Sodium hydrogen carbonate				
526	Calcium hydroxide				
338	Orthophosphoric acid		finished coco	essed as P_2O_5 in a and chocolate ducts	
334	L-Tartaric acid			shed cocoa and e products	
3.3 EMULSIFIERS		Maxim	um Level	Products	
471 Mo	ono- and di-glycerides of fatty acids			Products described under 2.1 to 2.4	
	cithins ycerol	GMP		"""	

442	Ammonium salts of phosphatidic acids	10 g/kg		"	"
476	Polyglycerol esters interesterified recinoleic acid	5 g/kg	15 g/kg	"	"
491	Sorbitan monostearate	10 g/kg	in combinati	on "	"
492	Sorbitan tristearate	10 g/kg		"	"
435	Polyoxyethylene (20) sorbitan monostearate	10 g/kg			
3.4	FLAVOURING AGENTS	Maxim	um Level	Products	
3.4.1	Natural flavours as defined in the Codex Alimentarius, and their synthetic equivalents, except those which would imitate natural chocolate or milk flavours ¹		MP	Products des under 2.1 to	
3.4.2	Vanillin	G	MP	Products des under 2.1 to	
3.4.3	Ethyl vanillin	GMP		Products described under 2.1 to 2.4	
3.5	SWEETENERS				
950	Acesulfame K	500 mg/kg		Products described under 2.1 to 2.4	
951	Aspartame	2 000	mg/kg	"	"
952	Cyclamic acid and its Na and Ca salts	500 mg/kg		"	"
954	Saccharin and its Na and Ca salts	500	mg/kg	"	"
957 420	Thaumatin				"
420 421	Sorbitol Mannitol			"	"
953	Isomalt	G	MP	"	"
965	Maltitol			"	"
966	Lactitol			"	"
967	Xylitol				
959	Neohesperidine dihydrochalcone	100	mg/kg	"	"
3.6	GLAZING AGENTS	Maxim	um Level	Produ	
414	Gum Arabic (Acacia gum)			Products des	
440	Pectin			under 2.1 to	2.4
901	Beeswax, white and yellow	G	MP	"	"
902	Candelilla wax			"	"
903	Carnauba wax			"	"

¹ Temporarily endorsed

904	Shellac		" "
3.7	ANTIOXIDANTS	Maximum Level	Products
304	Ascorbyl palmitate	200 mg/kg	Products described under 2.1.5. (White Chocolate)
319	Tertiary butylhydroquine	200 mg/kg	11
320	Butylated hydroxyanisole	200 mg/kg	"
321	Butylated hydroxytoluene	200 mg/kg	"
310	Propylgallate	200 mg/kg	"
307	α-Tocopherol	750 mg/kg	"
3.8	COLOURS (FOR DECORATION PURPOSE ONLY)	Maximum Level	Products
175	Gold	GMP	Products described
174	Silver	GMP	under 2.1 to 2.4
3.9	BULKING AGENTS	Maximum Level	Products
1200	Polydextrose A and N	GMP	Products described under 2.1 to 2.4
3.10	PROCESSING AID	Maximum Level	
Hexa	nne (62°C - 82°C)	1 mg/kg	
4	CONTAMINANTS	Maximum Level	
4.1	Arsenic (As)	0.5 mg/kg	
4.2	Copper (Cu)	20 mg/kg	
4.3	Lead (Pb)	1 mg/kg	

5 HYGIENE

5.1 It is recommended that the products covered by the provisions of this standard be prepared and handled in accordance with the appropriate sections of the *Recommended International Code of Practice - General Principles of Food Hygiene -* CAC/RCP 1-1969, Rev.3 - 1997, and other Codes of Practice recommended by the Codex Alimentarius Commission which are relevant to these products.

5.2 To the extent possible in good manufacturing practice, the products shall be free from objectionable matter.

5.3 When tested by appropriate methods of sampling and examination, the product :

- a) shall be free from microorganisms in amounts which may represent a hazard to health;
- b) shall be free from parasites which represent a hazard to health; and

c) shall not contain any substance originating from microorganisms in amount which represent a hazard to health.

6 LABELLING

In addition to the *Codex General Standard for the Labelling of Prepackaged Foods* — CODEX STAN 1-1985 - Rev. 1-1991, the following declarations shall be made:

6.1 NAME OF THE FOOD

6.1.1 Products described under Sections 2.1 to 2.4 of this Standard and complying with the appropriate requirements of the relevant section shall be designated according to the name listed in Section 2 under subsequent section and subject to the provisions under Section 6 of this Standard. The name of the product may be described with *'bittersweet chocolate*'', *'semi-sweet chocolate*'', and *"dark chocolate*'' as synonyms for *Chocolate*, as defined in Section 2.1.1,

6.1.2 [If the amount of any edible foodstuff added in accordance with the Section 2.1 of this Standard is equal or higher than 5% m/m of the weight of the product, a reference to these foodstuff or foodstuffs shall be included in the designation of the chocolate.

When sugar is fully substituted with sweeteners, an appropriate declaration should be included in the designation of the chocolate.

Example: X chocolate sweetened with aspartame.]

The use of vegetable fats in addition to Cocoa butter in accordance with the provisions of Section 2.1 shall be indicated on the label in association with the name and/or the representation of the product. The authorities having jurisdiction may prescribe the specific manner in which this declaration shall be made.

6.1.3 Cocoa Butter Confectionery / White Chocolate

Product described under Section 2.1.5 of this Standard and complying with the appropriate requirements of this Section shall be designated '*Cocoa Butter Confectionery*''. In those countries where the name "*White Chocolate*" is widely used, the use of this name is permitted.

6.1.4 Filled Chocolate

6.1.4.1 Products described under Section 2.3. shall be designated "X *Filled Chocolate*", "*Chocolate with X Filling*" or "*Chocolate with X centre*" where "X" is descriptive of the nature of the filling.

6.1.4.2 The type of chocolate used in the external coating may be specified, whereby the designations used shall be the same as stated under Section 6.1.1 of this Standard.

6.1.4.3 An appropriate statement shall inform the consumer about the nature of the centre.

6.1.5 A Chocolate or Praline (Bonbon Chocolat or Praline)

Products in a single mouthful size described under Section 2.4.1 of this Standard shall be designated "A *Chocolate*" or "*Praline*".

6.1.6 Assorted Chocolates

Where the products described under Section 2.1, 2.2, 2.3 or 2.4, with exception of chocolate a la taza, are sold in assortments, the product name may be replaced by the words "Assorted Chocolates" or "Assorted filled Chocolates", "Assorted Vermicelli", etc. In that case, there shall be a single list of ingredients for all the products in the assortment or alternatively lists of ingredients by products.

6.1.7 Other Information Required

6.1.7.1 Any characterizing flavour, other than chocolate flavour shall be in the designation of the product.

6.1.7.2 Ingredients which are especially aromatic and characterize the product shall form part of the name of the product (e.g. Mocca Chocolate).

6.1.8 [Use of the Term Chocolate

Products not defined under this Standard, and where the taste is derived from non-fat cocoa solids, can carry the term "chocolate" in their designations in accordance with custom and this to designate other products which cannot be confused with those defined in this Standard.]

6.2 [DECLARATION OF MINIMUM COCOA CONTENT

Products described under Section 2.1 of this Standard shall carry a declaration of cocoa solids except that governments of countries in which different names are used to differentiate the products may allow for no declaration. For the purpose of this declaration, the percentages declared shall be made after the deduction of the other permitted edible foodstuffs.]

6.3 [NET CONTENTS

Small units of up to 50 g may be exempted from a declaration of net weight on the label.]

6.4 LABELLING OF NON-RETAIL CONTAINERS

Information required in Section 6 of this Standard and Section 4 of the Codex General Standard for the Labelling of Prepackaged Foods shall be given either on the container or in accompanying documents, except that the name of the product, lot identification, and the name and address of the manufacturer or packer shall appear on the container. However, lot identification, and the name and address of the manufacturer or packer may be replaced by an identification mark provided that such a mark is clearly identifiable with the accompanying documents.

7 METHODS OF ANALYSIS

7.1 DETERMINATION OF ARSENIC

According to AOAC 952.13 (Codex general method).

7.2 DETERMINATION OF CENTRE AND COATING IN COMPOSITE AND FILLED CHOCOLATE

All methods approved for the chocolate type used for the coating and those approved for the type of centre concerned.

7.3 DETERMINATION OF COCOA BUTTER

According to AOAC 963.15 or IOCCC 14-1972.

7.4 DETERMINATION OF COPPER

According to AOAC 971.20 (Codex general method) or AOAC 960.40 (Codex general method).

7.5 DETERMINATION OF FAT-FREE COCOA SOLID

According to AOAC 931.05.

7.6 DETERMINATION OF FAT-FREE MILK SOLIDS

According to IOCCC 17-1973 or AOAC 939.02.

7.7 DETERMINATION OF LEAD

According to AOAC 986.15 (Codex general method).

7.8 DETERMINATION OF MILK FAT

According to IOCCC 8i (1962) or AOAC 945.34, 925.41B, 920.80.

7.9 DETERMINATION OF MOISTURE

According to IOCCC 26-1988 or AOAC 977.04. (Karl Fischer method); or AOAC 931.04 or IOCCC 1-1952 (gravimetry).

7.10 DETERMINATION OF TOTAL FAT

According to AOAC 963.15.

7.11 DETERMINATION OF VEGETABLE FAT

To be developed.

APPENDIX VI

LIST OF FOOD ADDITIVES PROPOSED BY THE DELEGATION OF JAPAN FOR USE IN COCOA POWDERS (COCOAS) AND DRY MIXTURES OF COCOA AND SUGARS, AND RELATED PRODUCTS:

	Emulsifier	Maximum level
491 492 493 494 495	Sorbitan Monostearate Sorbitan Tristearate Sorbitan Monolaurate Sorbitan Monooleate Sorbitan Monopalmitate	2 g/kg in combination in finished
477 475	Propylene Glycol Esters Of Fatty Acids Polyglycerol Esters Of Fatty Acids	5 g/kg in finished products 5 g/kg in finished products
412 418 466	Guar Gum Gellan Gum Sodium Carboxymethyl Cellulose	Limited by GMP in finished products
	STABILIZER	
400 407 410 413 414 415 416 417 460	Alginic Acid Carrageenan and its Na, N, NH4 salts Carob Bean Gum, Locust Bean Gum Tragacanth Gum Gum Arabic, Arabic Gum, Acacia Gum Xanthan Gum Karaya Gum Tara Gum Cellulose THICKENER	Limited by GMP in finished products
	Modified Starches	
1400 1401 1402 1403 1404 1405	Dextrins, Roasted Starch White And Yellow Acid-Treated Starch Alkaline Treated Starch Bleached Starch Oxidized Starch Starches, Enzyme-treated	Limited by GMP in finished products